

Make Your Publications Visible.

A Service of



Leibniz-Informationszentrum Wirtschaft Leibniz Information Centre

Yekani, Babalo; Ngcamu, Bethuel Sibongiseni; Pillay, Sareesha

Article

Monitoring and evaluation strategies: lessons learned from COVID-19 pandemic

Cogent Business & Management

Provided in Cooperation with:

Taylor & Francis Group

Suggested Citation: Yekani, Babalo; Ngcamu, Bethuel Sibongiseni; Pillay, Sareesha (2024): Monitoring and evaluation strategies: lessons learned from COVID-19 pandemic, Cogent Business & Management, ISSN 2331-1975, Taylor & Francis, Abingdon, Vol. 11, Iss. 1, pp. 1-18, https://doi.org/10.1080/23311975.2024.2353201

This Version is available at: https://hdl.handle.net/10419/326284

Standard-Nutzungsbedingungen:

Die Dokumente auf EconStor dürfen zu eigenen wissenschaftlichen Zwecken und zum Privatgebrauch gespeichert und kopiert werden.

Sie dürfen die Dokumente nicht für öffentliche oder kommerzielle Zwecke vervielfältigen, öffentlich ausstellen, öffentlich zugänglich machen, vertreiben oder anderweitig nutzen.

Sofern die Verfasser die Dokumente unter Open-Content-Lizenzen (insbesondere CC-Lizenzen) zur Verfügung gestellt haben sollten, gelten abweichend von diesen Nutzungsbedingungen die in der dort genannten Lizenz gewährten Nutzungsrechte.



https://creativecommons.org/licenses/by/4.0/

Terms of use:

Documents in EconStor may be saved and copied for your personal and scholarly purposes.

You are not to copy documents for public or commercial purposes, to exhibit the documents publicly, to make them publicly available on the internet, or to distribute or otherwise use the documents in public.

If the documents have been made available under an Open Content Licence (especially Creative Commons Licences), you may exercise further usage rights as specified in the indicated licence.





Cogent Business & Management



ISSN: 2331-1975 (Online) Journal homepage: www.tandfonline.com/journals/oabm20

Monitoring and evaluation strategies: lessons learned from COVID-19 pandemic

Babalo Yekani, Bethuel Sibongiseni Ngcamu & Sareesha Pillay

To cite this article: Babalo Yekani, Bethuel Sibongiseni Ngcamu & Sareesha Pillay (2024) Monitoring and evaluation strategies: lessons learned from COVID-19 pandemic, Cogent Business & Management, 11:1, 2353201, DOI: 10.1080/23311975.2024.2353201

To link to this article: https://doi.org/10.1080/23311975.2024.2353201

9	© 2024 The Author(s). Published by Informa UK Limited, trading as Taylor & Francis Group
	Published online: 14 May 2024.
	Submit your article to this journal 🗹
hil	Article views: 1103
Q	View related articles 🗗
CrossMark	View Crossmark data ☑



MANAGEMENT | RESEARCH ARTICLE



Monitoring and evaluation strategies: lessons learned from COVID-19 pandemic

Babalo Yekania, Bethuel Sibongiseni Ngcamub n and Sareesha Pillayc

^aDepartment of Management and Governance, Faculty of Business Sciences, Walter Sisulu University, East London, South Africa; bDepartment of Public Administration and Management, Faculty of Public and Operations Management, University of South Africa, Pretoria, South Africa; 'Public Management and Leadership Department, Nelson Mandela University, Port Elizabeth, South Africa

ABSTRACT

The COVID-19 pandemic presents unique challenges that necessitate a new approach to monitoring and evaluation (M&E) in local government. The culture of poor performance, high incompetency levels among employees and financial mismanagement widely publicised in popular media in South Africa – prompted this study to investigate the effectiveness of M&E strategies. The study attempts to determine the effective M&E strategies in supporting two municipalities' response that extends beyond COVID-19. A transformative mixed methods approach was used for data collection and analysis of data. The targeted respondents were solely responsible for COVID-19 activities at the Nelson Mandela Bay Municipality and the Buffalo City Metropolitan Municipality. In-depth interviews were utilised to collect data from 13 senior municipal managers. A total of 191 questionnaires were despatched to municipal employees and 186 were returned, giving a 97% response rate. The findings depict ineffective M&E tools and inadequate organisational support to sustain best practices on municipal governance. The findings unearthed a linear relationship between the municipalities' leadership failure to implement a performance management system among its workforce and an exponential increase in COVID-19 cases and fatalities. This study attempts to make a valuable contribution in providing municipal employees with tailored work experience that will be relevant for future disasters similar to COVID-19. adding value to knowledge.

ARTICLE HISTORY

Received 2 March 2022 Revised 13 March 2024 Accepted 29 April 2024

KEYWORDS

Digital management; implementation; intervention; monitoring and evaluation: municipalities; performance management

REVIEWING EDITOR

Huifen (Helen) Cai, Middlesex University Business School, UK

SUBJECTS

Economics; Business, Management and Accounting; Urban Studies

1. Introduction

In response to the challenges posed by the COVID-19 pandemic, concerns have been raised regarding the state of South African municipalities including corruption, white-collar crime, ineffective M & E systems, and governance violations. Notably, the latter governance pitfalls have pinpointed by a plethora of researchers (Mantzaris & Ngcamu, 2020b; Mlambo & Masuku, 2020; Sibanda et al., 2020). The culture of poor performance, inadequate workforce, and financial mismanagement at a municipal level further highlight the significance of implementing robust M&E systems (Dlamini & Migiro, 2016; Govender, 2013; Kariuki & Reddy, 2017; Levin, 2017; Mello, 2018; Motingoe & Van der Waldt, 2013). In light of these concerns, this study aims to explore effective M&E strategies that can bolster municipal responses to the COVID-19 pandemic, transcending the immediate crisis. The main research question guiding this investigation is, 'Are M&E strategies in South African municipalities adequately responsive to the impact of the COVID-19 pandemic?'

This study investigates and put into perspective the COVID-19 pandemic and its implications for service availability. As a result, causes and pressures such as resource constraints are explored as significant barriers to the effective delivery of municipal services (Seasons, 2003). Many complicated social, economic and environmental challenges have manifested in municipal government during COVID-19 and need to be addressed (Chaplowe & Cousins, 2016). The global impact of the COVID-19 pandemic has been profound, affecting life in various ways (Kadel et al., 2020). Governments worldwide grapple with unprecedented uncertainty and challenging trade-offs related to health, the economy, and social issues. This necessitates a new perspective on M&E practices, suggesting their integration into government strategies and emphasizing the need for adaptability in response to the unique challenges posed by the pandemic (Rosenberg & Kotschy, 2020; Seasons, 2003).

There is a pressing need for enhanced M&E support for South African municipalities, especially in complex, dynamic development contexts and emergencies like the COVID-19 pandemic (Rosenberg & Kotschy, 2020). However, the ineffective application of M&E tools in the Nelson Mandela Bay Municipality (NMBM) and the Buffalo City Metropolitan Municipality (BCCM) hinders their ability to manage effectively and meet service delivery mandates during COVID-19 (South African Local Government Association [SALGA], 2020). This study's findings have the potential to contribute valuable insights, informing broader crisis management applications and enhancing the adaptability and effectiveness of future M&E strategies in diverse emergency situations.

This article is structured as follows: Section 2 presents the relevance of the study and the conceptualisation of M&E as it relates to COVID-19. Section 3 reviews the literature systematically. Section 4 describes the design and methods with Section 5 presents, analyse and discuss the results for the study. Lastly, Section 6 provides the major highlights, conclusions and recommendations for the study.

2. Relevance of the research

This article illustrates the role of M&E in these two selected municipalities in order to enforce initiatives to protect citizens from COVID-19 and to plan for subsequent reactions to improve the national or municipal response against future pandemics (Awasthi, 2020). In dealing with the COVID-19 pandemic, municipalities required effective M&E systems more than ever in order to design effective policies (Evans, 2020). As a result, municipal employees need to enhance M&E tools. This includes performance management systems, financial management systems, stakeholder management systems, and data and information management. This is in order to maintain essential services in unprecedented times of crisis in the fight against the pandemic and to ensure community solidarity (SALGA, 2020). The two selected municipalities under consideration have encountered various crises including instances of severe droughts and water shortages, which also contributing to major socio-economic impact (Mahlalela et al., 2020). This situation prompts the necessity for a comparative analysis of COVID-19 with other types of crises to deepen a comprehension of how M&E strategies can be tailored to suit different scenarios.

The impact of COVID-19 will be felt far beyond the pandemic and municipalities have to play a supportive role and move with unprecedented ability to adapt to national and provincial government's changing regulations. The Section 11(3) of the Municipal Systems Act of 2000 highlights the significance of M&E in local government. This section emphasizes the role of M&E in overseeing and regulating municipal services, assessing the impact and effectiveness of various services, policies, programmes, or plans, and the establishment and implementation of performance management systems. This policy guideline does not offer any exemptions or justifications for municipalities to refrain from implementing M&E systems under any circumstances. Hence the human resource management should have continued as usual with revised performance agreements despite these circumstances. This proves that effective M&E strategies are the backbone of successful epidemic elimination programmes, providing countries with important systematic guidelines about when initiatives are required, in order to document the effects and identify where and how a pandemic can be prevented (Kusek & Rist, 2004). Consequently, Fraser-Moleketi (2020) posits that M&E now needs to be located in the broader discourse on sustainable development and the achievement of system-wide development outcomes. There is little or no M&E literature on COVID-19 in particular in the South African local government. This provides an opportune gap which the current study aims to fill. In bridging this gap, the current article seeks to identify effective M&E tools, explores the gaps and limitations in its assessment and ultimately suggests strategies to assess effectiveness rigorously. Accordingly, the study attempts to determine effective M&E strategies in supporting municipalities' response to COVID-19 that extends beyond this phenomenon.

3. Literature review

3.1. A theoretical and conceptual framework: monitoring and evaluation of COVID-19

Monitoring and evaluation are the key concepts to be studied, with a special focus on COVID-19 in local government. To begin with, a monitoring exercise is said to be a continuing function that uses systematic collection of data on specified indicators to provide management and main stakeholders of an ongoing development intervention with indications of the extent of progress and achievement of objectives and progress in the use of allocated funds (Kanyamuna et al., 2019). Meanwhile, evaluation systematically and objectively assesses a policy, programme or project which is an ongoing or completed process, from planning, design, execution and results (Ile et al., 2019). Furthermore, the coronavirus disease 2019 is a newly discovered disease which is an infectious virus (World Health Organization, 2020) that has caused radical instability to governments around the world (Organisation for Economic Co-operation and Development [OECD], 2020). The study interlinks these variables in order to achieve a successful response to the COVID-19 pandemic and to adopt effective M&E strategies in supporting the municipalities. As a result, this study sought to test the null hypothesis that there are no effective M&E strategies and no organisational support to sustain best practice in the two selected municipalities during COVID-19.

Monitoring and evaluation tools are integral parts of the performance analysis of organisations, in addition to the design and implementation phases of plans and projects in many fields (including the COVID-19 pandemic) (Gaskin-Reyes, 2016). This is consistent with the WHO's COVID-19 Strategic Preparedness and Response Plan framework which aims to assess national plans, country and regional response performance and progress (World Health Organization, 2020).

Monitoring and evaluation is used for a variety of goals, including discovering what works, improving and redesigning programme approaches, and determining cause and effect (Kettner et al., 2015). When M&E is successfully implemented, it entails reporting downwards to local communities, and upwards to national government for the purpose of accountability and credibility (Bell & Aggleton, 2016). The relevance of M&E to COVID-19 is that information collecting as part of monitoring permits the supply of information. This helps organisations and stakeholders to access the level of success and achieve goals as well as make progress in the utilisation of available funds (Ile et al., 2019). Evaluation, meanwhile, emphasises the significance of the strategy or initiative in determining the relevance of the goals, the feasibility of the design and implementation, and the sustainability of the outcomes (Awasthi, 2020). Effective M&E systems can promote recovery by ensuring that funds are spent wisely and effectively (Awasthi, 2020). In view of the above, this article sought to conceptualise the effectiveness of M&E strategies during COVID-19 in the two selected municipalities.

3.2. Governance and performance management: studying monitoring and evaluation strategies during COVID-19

The municipalities' leadership failure to implement M&E among its workforce is a significant challenge that has had an impact on municipalities' performance (Ngqobe et al., 2021). This problem could be minimised if the two municipalities adopt M&E strategies as strong governance management initiatives and services (Kanyamuna & Phiri, 2019). There has been an increase in awareness creation of M&E in public and private sectors and organisations (Mkama, 2017). Today, the Sustainable Developmental Goals (SDGs) have advocated for stronger M&E systems at the country level, to be used as tools to track performance. This would be for the purposes of accountability, feedback and learning to improve current and future development interventions (Kanyamuna et al., 2020). In light of the need to meet the SDGs and the enormity of demand, the lack of M&E capacity presents a potential obstacle in some countries (Evans, 2020). The use of M&E tools is important in tracking and judging the progress of any project (Mleke, 2020). It is evident that effective M&E is pivotal for good governance and public resource management because it promotes transparency, accountability and a culture of performance (African Development Bank, 2009). A developing country such as South Africa will need to innovate and learn how to continue with vital practices such as M&E under the new restrictive conditions (Fraser-Moleketi, 2020).

In view of the M&E strategies used in municipalities during COVID-19, this study focuses on the performance management system (PMS) as a tool that informs organisational performance for responsible and responsive M&E (Draai & Zazi, 2021). The change in the implementation of the performance management policy indicates that COVID-19 affected the normal way of conducting performance assessments (Pietersen, 2020). The PMS in the public sector is designed to integrate the performance of employees and the organisation (Malefane, 2009). As a result, this author argues for the necessity of ensuring that individual and team performance is assessed for its alignment with the organisational strategy (Malefane, 2009). It became a challenge to conduct performance assessments during COVID-19 because people were required to work from home (Pietersen, 2020). The lack of completed performance assessments indicates that the system is reliant on human contact and the workplace setting (Prodanova & Kocarev, 2021). This suggests that if working from home is to be a future practice, it will be important for the public sector to innovate and develop electronic systems for performance management and assessment (Pietersen, 2020). Overall, the COVID-19 pandemic affected the two municipalities, particularly the conditions of service (Khambule, 2020). By extension, service delivery was also affected as some services were suspended or scaled down as priority was given to essential services (Mlambo & Masuku, 2020). These authors state that municipalities have been required to adopt human and technological shifts in service conditions to ensure integrated human resource practices and the minimal disruption of municipal services (Mlambo & Masuku, 2020).

These shifts could include policies on redefining the workplace, the digital management of performance management, improved leave management linked with working from home and properly managed collective bargaining processes (Pietersen, 2020). This author suggests that in doing so, the public sector could benefit from an integrated strategic human resource system and improved employee performance. Pietersen (2020, p. 446) argues that a paradigm shift is required for sustainable change, even though the temporary solutions regarding conditions of service during the pandemic were relatively well managed. Pietersen (2020) further asserts that municipalities must continue the momentum of considering swiftness when dealing with the conditions of service as this can translate into effective and efficient service delivery. More importantly, the lessons learnt from the COVID-19 challenges should be turned into success stories (Sibanda et al., 2020). The reviewed literature above on M&E as a responsive tool to the impact of the COVID-19 pandemic in municipalities illustrates circumstantial information that is not supported or validated by empirical evidence. This suggests a serious gap in the literature, which requires further exploration by researchers in order to present conclusive evidence.

3.2.1. Transferable strategies and skills

South African municipalities under the pandemic's pressure have transitioned to remote monitoring, a shift emphasized by the need for adjustments in M&E plans and approaches (Pietersen, 2020). This shift not only responded to the immediate challenges posed by COVID-19 but also equipped municipalities with transferable skills and strategies to navigate future uncertainties and changes in the broader landscape. Remote monitoring, although not new, has gained prominence during the pandemic, particularly in conflict areas and crisis situations where physical access to project sites is challenging (Dube et al., 2021). The shift towards remote M&E while beneficial presents challenges and requires municipalities to empower employees with technology and cover connectivity costs. However, this intervention might increase the M&E budget involving substantial expenditures on tablets and smartphones for remote data collection (Ile et al., 2019). Subsequently, the capacity to implement M&E strategies requires to integrate skill development into overall institutional strategies. Identifying skill gaps and implementing targeted capacity-building interventions are key steps in enhancing the effectiveness of M&E in municipalities. Additionally, embedding M&E roles in job descriptions and linking individual performance to the institutional M&E system is crucial. Furthermore, recognition frameworks acknowledging M&E achievements and strategic recruitment and training initiatives became essential components to attract and retain specialized M&E skills (Ile et al., 2019). Furthermore, the scope of M&E extends beyond operational aspects such as strategic elements like policy implementation, stakeholder engagement, and communication strategies.

4. Research methodology

The study employed mixed methods approach within a convergent core design and the results were merged. It was necessary to use mixed-methods design; because it combines both the qualitative and the quantitative data collection analyses and interpretation (Deji, 2019). In-depth interview questions were utilised to collect data and 22 municipal senior managers (heads of directorates/departments) were

targeted for one-on-one interviews. The researchers used their judgement to select the municipal departments that have continued to provide essential services to communities during the COVID-19 lockdown. This includes waste removal, water, sanitation, electricity and disaster management, and safety and wellness. Out of 22 municipal senior managers, only 13 managers were interviewed. The researchers were able to reach the degree of saturation (Guest et al., 2020). The questionnaires were structured according to the specific research objectives and distributed to the municipal employees in different employment categories. The questionnaires were written in English, considering it is a widely used language in municipalities. The structured questionnaires were created using a 1-to-5 Likert scale, with 1 indicating strongly disagree, 2 indicating disagree, 3 indicating undecided, 4 indicating agree and 5 indicating strongly agree. A total sample of n=191 was sampled, according to the Sekaran (2002) sampling table.

The targeted respondents were solely responsible for COVID-19 activities or operations. In total, 191 questionnaires were despatched and 186 were returned, which gave a 97% response rate. The International Business Machines Statistical Package for the Social Sciences (IBM SPSS) version 27.0 was used to analyse the quantitative data (Bandalos, 2018) and NVIVO version 12 to analyse the qualitative data. The Cronbach alpha values were 0.888 and the total number of items was 13 (.888 N = 13), used to determine the reliability of the individual dimensions. The reliability coefficient indicated a high degree of acceptability, in addition to consistent scoring for the different categories of this research (Taber, 2018).

The chi-square (chi-square p-value < 0.001) was used to tests the null hypothesis. The chi-square had a value of zero since the fit was perfect (Moore et al., 2013). The p-value of 0.001 was less than the level of significant difference between the number of respondents who agreed and those who disagreed with a statement. The null hypothesis was rejected and the alternative was accepted since the p-value is smaller than 0.5 (level of significance). In addition, the factor analysis was done to discover patterns and associations among variables (Babbie & Mouton, 2001). The factor loading was above 0.5 and was effectively measured along the various components. The standard deviation was made to discover how respondents answering questions varied or deviated from the mean (Quinn & Keough, 2002).

A thematic analysis was used to analyse the qualitative data from the interview schedules and also the one open-ended question from the questionnaires. This eventually provided an opportunity to identify themes used for the discussions in this study (Charmaz, 2009). The research instruments were informed by the research study's objectives and attempts to gather elaborative answers to the research questions. Interviews and questionnaires were crafted using research tools theory (Quad, 2016).

4.1. Ethical guidelines

The researchers were granted ethical clearance by the Nelson Mandela University's Research Ethics Committee to conduct research at the NMBM and the BCMM. During the collection of data, the researchers was guided by the Protection of Personal Information Act, 2013 (Act No. 41 of 2013) (Republic of South Africa, 2013) and maintained the anonymity and confidentiality of the respondents throughout the study. This Act aims to protect the personal information processed by public and private bodies, including local authorities (Netshakhuma, 2020).

5. Results and discussion

The research objectives were to determine effective M&E strategies in supporting the municipalities' response to COVID-19 that extends beyond this phenomenon. The dimension of investigating the effectiveness of M&E strategies during COVID-19 was established. This aspect deals with the responses of employees when asked to respond to opinion statements that sought to assess the effectiveness of M&E strategies during the pandemic. Table 1 summarises the scoring patterns. Literature was also used to support and validate the study's findings and interpretations of results to provide additional information to be compared with the existing data (Fouché et al., 2022).

Table 1 reveals that 41% of the participants agreed that the COVID-19 pandemic was handled well in these two municipalities, whereas 34% of the participants disagreed and 24% were undecided. The study shows the negative finding that employees have poor faith in these municipalities' ability to handle COVID-19 effectively, with a disproportionately high percentage of 34% employees who disagreed and 24% of employees who were

 Table 1. The effectiveness of monitoring and evaluation strategies during COVID-19.

ימור בווכבוולים כו וויסוונסווון מוומ בימוממיסון פמיניקובי ממווון כי ביום ופי								
	Strongly				Strongly		Standard	
Sub-dimension	disagree	Disagree	Undecided	Agree	agree	Mean	deviation	Chi-square test
COVID-19 was handled well	12	22	24	33	8	3,0269	1,17373	< 0.001
Employees clearly understand the monitoring and evaluation system	7	20	20	43	10	3,3011	1,10316	< 0.001
Communication of the lockdown rules and regulations was clear	7	14	11	48	21	3,6290	1,15174	< 0.001
There are adequate policies to support the monitoring and evaluation system	7	22	56	33	13	3,2527	1,13214	< 0.001
Employees' work was monitored during COVID-19	13	17	21	38	11	3,1505	1,22537	< 0.001
The performance management system is used as one of the monitoring and evaluation tools	6	18	29	33	11	3,1989	1,12860	< 0.001
The stakeholders were involved in the COVID-19 response plans	80	12	27	42	11	3,3602	1,07267	< 0.001
All the employees have a performance agreement	18	24	31	21	5	2,7097	1,14920	< 0.001
The performance agreement specifies the objectives and targets that should be met during COVID-19	15	20	28	30	7	2,9409	1,17715	< 0.001
Time frames were set continuously to monitor my work during the COVID-19 crisis	20	23	18	29	6	2,8333	1,29830	0.001
My work is evaluated whenever there is an early indication of poor performance	17	17	79	36	5	2,9570	1,18014	< 0.001
Monitoring and evaluation are concepts for the top managers only	16	20	22	29	13	3,0484	1,28729	0.008
Monitoring and evaluation are very important	8	4	14	32	46	3,9839	1,18311	< 0.001
Source: authors.								

undecided. The empirical finding is no different from a previous investigation conducted almost a year ago, as reported by National Treasury (2020). This finding is consistent with a report by National Treasury (2020) that confirmed a lack of accountability in many municipalities and an inability to effectively manage key projects and programmes (the COVID-19 pandemic cannot be excluded).

A theme emerged regarding the managing of COVID-19 pandemic in these two municipalities. The study findings concurred with the finding on the perceptions held by municipal employees. A female interviewee in Economic Development Department at the NMBM expressed that:

I will begin with the M&E practices which I would say, its 50% good and 50% bad. As for compliance with COVID-19 pandemic protocols, I would rate it at about 60% because there are several aspects that have not been done correctly.

A male senior manager in the Spatial Development Department from BCMM had this to say:

Ah ... I would say average. In this sense, COVID-19 has different aspects. There is an issue of personal protective equipment; as far as that is concerned I would say that the municipality has handled that one well because all buildings have [been] supplied with sanitiser, gloves, masks, thermometers, all of that and there are regular checks on people who enter the buildings, so that is practised well. Then we come to the aspect of comorbidities; staff with comorbidities have not been handled well because we have a situation where people [have been] away from work for 12 months since the lockdown level 5 and they have not been back. Yet you see those people in malls, you see them attending functions, even funerals, and post them on social media pages but they do not come to work. And you have a situation where even people who are getting the car allowances are still receiving car allowances whereas they [cars] are not driven for work purposes for a year now; from that point of view, I would say the municipality has not handled the crisis well.

This finding is another one of interest – that managing municipal employees who had health complications or comorbidities were not handled well during COVID-19. This finding resonates with the National Treasury (2020) who established that municipal employees received benefits and grants to which they were not entitled, due to a lack of validation, integration and data sharing across departments. This qualitative empirical finding affirms that risk factors threaten accountability in both metropolitan areas. The competence of municipal officials in management, and accountability is key to addressing and reducing the effects of COVID-19 on citizens (Ball, 2020). This finding confirms that during pandemics like COVID-19, there is a critical demand for stronger, more effective M&E systems, particularly at the municipal level (Fraser & Morkel, 2020; Rosenberg & Kotschy, 2020).

Table 1 indicates that the highest percentage - 53% of respondents - agreed that employees in these municipalities understand the M&E system. This is against 27% who disagreed, with only 20% who were undecided. Despite there being a higher percentage of respondents who agreed that employees clearly understand the M&E system, there is a concern about those who disagreed and others who were undecided. This finding requires further consideration and trust building. This can be done by ensuring that every employee in these municipalities is given the necessary support to clearly understand the M&E system. This finding is contrary to the study conducted by Awasthi (2020) who suggests that in the current situation, it is important to emphasise the role of M&E. This is in order to enforce initiatives to protect citizens from COVID-19 and to plan for subsequent reactions to improve the national or municipal response during similar future pandemics. Furthermore, with just 53% of respondents agreeing to this sub-dimension, this reveals that there are gaps and loopholes in these municipalities. This finding confirms the findings of Eresia-Eke and Boadu (2019), who discussed the challenges and gaps in M&E systems, arguing that the absence of the prerequisites for M&E, even within organisations that already have an M&E system, may have a detrimental impact on the performance of the institutions.

The mean score values displayed in the sub-dimension that employees clearly understand the evaluation of M&E systems reveal that on a scale from 1 to 5, the respondents were between 3,3011 and the standard deviation was 1,10316. Again, this shows that a high proportion of municipal employees varied from agreeing to being undecided on statements relating to this sub-dimension. These averages reflect the current situation at the municipalities concerned - that not all employees clearly understand the M&E system. It is worth emphasising the importance and the role of M&E in order for these municipalities to improve their performance in managing and controlling the COVID-19 pandemic.

Another emerging theme regards the understand M&E systems in municipalities. A male senior manager in the Disaster Management Department at the BCMM said: 'No, it is only a few, I think that's one of the challenges because there has not been a roll-out on educating employees on how to use M&E systems'. The same response a senior manager from the Disaster Management Department at the NMBM explained: 'On junior level, I would say there could be some improvement where people could be better informed as to M&E, and that this should be rolled out and implemented on a more junior level as well, not just at senior management level.'

This finding implies that M&E is understood better at the senior management level and is not fully implemented for employees across job levels. This finding resonates with the plethora of empirical studies conducted in South Africa. These studies highlight challenges facing M&E systems in municipalities, which are experiencing moderate and significant problems with M&E implementation (Mapitsa & Korth, 2017; Matsiliza, 2019; Mthethwa & Jili, 2016; Ngwakwe, 2020).

Table 1 shows that there is a disproportionately high percentage – 69% of respondents – who agreed that the communication regarding the lockdown rules and regulations was clear, with only 21% disagreeing with the statement. A total of 11% were undecided. This finding is consistent with the suggestion of earlier work as quoted from the literature – that the successful implementation of COVID-19 regulations requires efficient governance, a clear understanding of government policies and continuous M&E (Mlambo & Masuku, 2020).

The sub-dimension regarding the communication of the lockdown rules and regulations were clear and reflect mean score values between 3,6290 and a standard deviation of 1,15174. These results indicate that the respondents were in favour of the statement in this sub-dimension. However, a high proportion of municipal employees who ranged from 'agreed' to 'undecided' on statements relating to this sub-dimension cannot be ignored. This average reveals that there is a gap and a need to ensure that the communication of lockdown rules and regulations reaches every employee.

Although the majority of respondents agreed that communication of the lockdown rules and regulations were clear, there is a concern about whether the municipal employees have been complying with these rules and regulations. One female senior manager in the Economic Development Department at the NMBM revealed an interesting finding, stating that:

To be honest with you, compliance – it's a bit of a problem because even now when you walked in, I'm sure you will tell me what happened when you walked in. There are two people sitting in a small security container. Sometimes when it's hot, they're outside, but when it's cold, the security stays there. That was why you find that most of the people that got COVID-19 in the municipality are securities. Those are the people who worry me sometimes. The cleaners as well, because our cleaners are using change rooms.

This finding is contrary to national studies which have shown conclusively that in instances such as the COVID-19 pandemic, municipalities have to play a key role in protecting communities, particularly those that are vulnerable. This includes careful planning and implementation and the rigorous application of disaster preparedness, mitigation and prevention (Corburn et al., 2020). The finding is contrary to several COVID-19 national regulations, relating to section 3(1) of the Disaster Management Act, 2002 (Act No. 57 of 2002) and section 27(1) of the Act that declares a national state of disaster (Republic of South Africa, 2002). This Act is considered necessary for the provision of relief and assistance, and the protection of the public and their property against the effects of the COVID-19 pandemic (Republic of South Africa, 2020). The finding above is particularly visible at these municipalities and is consistent with the previous study conducted by Sibanda et al. (2020). These authors analysed the effectiveness of institutional arrangements and governance, municipal enforcement agencies, the efficiency of response plans to the pandemic, and municipal operations. As a result, these authors' findings pinpoint the absence of M&E as having led to non-compliance trends in South African municipalities during COVID-19 (Sibanda et al., 2020).

Table 1 portrays a total percentage of 46% of the respondents who confirmed that the policies were adequate in supporting M&E systems, followed closely by those who had a different perception: 29% disagreed and 26% were undecided. For those who remained undecided, the impression is that they might not be sure about the existence or non-existence of municipal policies. This finding is consistent with the study of Fraser and Morkel (2020) who concur that, although the system of M&E may be in place in South Africa, the levels of performance of government, as evaluated by citizens, remains wanting.

The mean score values displayed in the sub-dimension indicate that the policies were adequate in supporting M&E systems and reflect that on a scale from 1 to 5, the respondents were between 3,2527 and the standard deviation was 1,13214. This indicates that a high proportion of municipal employees ranged from 'agreed' to 'undecided' on statements relating to this sub-dimension. These averages reflect that some municipal employees are in the dark concerning the municipal policies that support M&E. The current research study supports the findings of Gxabuza and Nzewi (2021) - that M&E in these municipalities seems to be at the initial stage and has not been fully established, which is still a problem. This finding is contrary to the study conducted by Evans (2020), who argued that now more than ever, each municipal government needs robust M&E systems in order to design effective policies.

Another noteworthy theme that emerged concerns the adequate policies to support the M&E systems. A contrary point of view was expressed by a female senior manager at the BCMM in the M&E Department, who stated:

The M&E still [has] a challenge and the moment senior managers don't see it as their responsibility and [are] not implementing what is really outlined in the framework, then there's a challenge. So there is a gap. It's not really - I don't want to say M&E [is] not effective - I will rather say M&E ... it's not fully implemented. I think nobody really cares. Nobody sees it as important.

Taking these findings into consideration, this result is contrary to a number of studies which concluded that effective M&E policies need to be adopted and implemented to promote good governance in local municipalities (Mlambo & Masuku, 2020; Munzhedzi & Makwembere, 2019). The successful implementation of COVID-19 regulations requires efficient governance, a clear understanding of government policies, continuous M&E and amendments to a weak policy. Furthermore, the disparity in the analysis implies that there may be other policies that employees are unaware of. This is a matter that needs to be addressed in these municipalities in order to promote good performance. The issue of implementation of the framework needs significant change management. The officials responsible appear to be so used to the way things are and comfortable with the old ways of doing things. This empirical finding resonates with Eresia-Eke and Boadu (2019) recommendations that future research should take a critical look at the difficulties confronting government institutions as they attempt to address issues related to the absence of some of the prerequisites for effective M&E.

Table 1 shows a percentage of 49% of respondents who agreed that employees' work was monitored during COVID-19, against 30% who disagreed and 21% who were undecided. This finding implies that not all employees' work is monitored in these municipalities. This finding is opposed to the recent studies conducted by Kanyamuna et al. (2020), Mleke (2020) and Ile et al. (2019) who all concur that the main purpose of monitoring is to identify the progress and shortcomings and to recommend corrective measures to optimise desired results.

The sub-dimension of employees' work being monitored during COVID-19 reflects mean score values between 3,1505, while the standard deviation was 1,22537. This sub-dimension indicates that a high proportion of municipal employees ranged from 'agreed' to 'undecided' on the statement. This average reflects that employees' work was in fact not monitored during COVID-19.

Another theme was whether the work of employees was monitored during COVID-19. Negative findings, meanwhile, were also confirmed by a female senior manager from BCMM in the M&E Department who was interviewed qualitatively. She gave the following feedback when asked if employees' work was monitored during the pandemic:

Most of the reports, you find out the quality is compromised. Also the issue of people who are over 60 years, they were not here; it was a problem because you're not sure whether when the information you get is credible information or not. In some cases, most of the KPIs of the performance information was compromised because there was no movement because people were on lockdown, it was really bad. In terms of time frames, in terms of content, in terms of the personnel available, etc.

This finding has conclusively shown that instances such as the COVID-19 pandemic affected the normal way of conducting performance assessments (Pietersen, 2020). It became a challenge to monitor employees' work during COVID-19 because staff were required to work from home. It will be important for the municipalities to innovate and develop electronic systems to monitor employees' work even beyond COVID-19 as working from home may be a future practice (Mlambo & Masuku, 2020). Monitoring could include tracking performance, although it is unlikely to establish the reasons for good or poor performance (Ile et al., 2019). This paradigm shift is required for sustainable change, even though the temporary solutions for conditions of service during the pandemic were relatively well managed (Pietersen, 2020).

Table 1 shows a total of 44% of respondents who agreed that the PMS is used as one of the M&E tools, against 27% who disagreed. Only 29% were undecided. This is another highlight of the low percentage of respondents who agreed; when those who disagreed and those who were undecided are combined, they form a majority. This implies that the majority of municipality employees are in the dark concerning the PMS and this leaves a significant gap for improvement. This finding is contrary to the study of Malefane (2009, p. 143) who concluded that the PMS in the public sector is designed to integrate the performance of all employees and the organisation. Seemingly, the PMS in these municipalities has not been fully established.

The mean score values displayed in this sub-dimension – on the PMS being used as one of the M&E tools in these municipalities – reflect that on a scale from 1 to 5, the respondents were between 3,1989 and the standard deviation was 1,12860. This indicates that a high proportion of municipal employees ranged from 'agreed' to 'undecided' on statements relating to this sub-dimension. These averages reflect the discomforting situation at the municipalities concerned, as the PMS creates a climate that is not conducive to motivating employees to develop and achieve high standards of performance (Public Service Commission (PSC), 2008). Furthermore, it helps with the management and integration of the performance of institutions and that of their employees (Den Hartog et al., 2004).

According to the theme that PMS is used as M&E tools in municipalities. The study findings have also been confirmed by senior municipal managers who were interviewed qualitatively. They concurred with this finding on the perceptions held by municipal employees. A female senior manager in the M&E Department from NMBM concerned had this to say:

There is no framework from COGTA or whatever. There's nothing – [no] policy, no Act, there's no regulation that governs the performance information on levels below section 56 and 57 managers. Hence in all cities, there is no city that masters employee performance management. For example, in [the] M&E unit, I'm responsible up until section 56 or 57 managers. Then the Human Resource Department is supposed to instil employee performance management from general managers level going down. So there is that challenge.

The necessity for a PMS is to ensure that the alignment of individual and team performance is assessed for its agreement with the organisational strategy (Malefane, 2009, p. 143). The municipalities must continue the momentum of considered swift response when dealing with the conditions of service as this can translate into effective and efficient service delivery. In doing so, the municipalities could benefit from an integrated strategic human resource system and improved employee performance (Pietersen, 2020).

Table 1 shows a high percentage of 53% of respondents who agreed that the stakeholders were involved in the COVID-19 response plans. A total of 20% disagreed and only 27% were undecided. This finding is consistent with the study by Boadu and IIe (2017) who concur that to achieve results through M&E, stakeholder participation is essential because development results have to be defined from a stakeholder's perspective. In addition, the extent to which the results have or have not been achieved can only be established with the co-operation of the stakeholders.

Despite the higher percentage of respondents who agreed that the stakeholders were involved in the COVID-19 response plans, there is cause for concern because of those who disagreed or were undecided. The mean score values in the sub-dimension of the stakeholders being involved in the COVID-19 response plans in these municipalities showed that on a scale from 1 to 5, the respondents were between 3,3602 and the standard deviation was 1,07267. This indicates that a high proportion of municipal employees ranged from 'agreed' to 'undecided' on statements relating to this sub-dimension. Much room for improvement is reflected for this sub-dimension in terms of stakeholders being involved in the COVID-19 response plans. Stakeholders are best positioned to volunteer information about the COVID-19 pandemic and if the original problem of concern is being addressed by the developmental initiative (Xue et al., 2013).

Despite the majority of respondents in the quantitative findings agreeing on the theme that the stakeholders were involved in the COVID-19 response plans, there was a contrary view among senior

managers in these municipalities. A female senior manager in the Economic Development Department at NMBM confirmed the following:

Shame, that one has been non-existent and I don't want to lie. It has been non-existent because there's no way that people are having meetings. The only thing that we've managed to do is through social media, but this for me is a problem because it's only able to reach a certain portion of people. So in terms of the media, they've done a very good job. When it comes to road shows and loud hailers, that has also [been] done a lot. The municipality could have done more and we can still improve because remember ... this is a new field. It's a new field and we are still grappling.

This finding is shocking and is contrary to the study conducted by Kusek and Rist (2004). These researchers stated that the knowledge of stakeholders helps an organisation to determine what interests exist and how these can be addressed by the initiative and by the information generated by the M&E system. An understanding of stakeholders is therefore useful in determining results, monitoring and evaluating the extent that results are realised and establishing roles for stakeholders (Kusek & Rist, 2004) even during COVID-19. This is because many roles and responsibilities require high levels of commitment and only a buy-in on the part of stakeholders can guarantee success and sustainability (Ile et al., 2019).

Table 1 shows a high percentage of 42% of respondents who disagreed that all the employees have a performance agreement, followed by another high percentage of 36% who were undecided, with only 26% agreeing. This finding is consistent with the finding above on the sub-dimension that confirmed the negative perception of employees regarding PMS. It proves that PMSs are absent in these municipalities. As a result, there is no alignment or assessment of individual and team performance to ensure there is agreement with the organisational strategy (Malefane, 2009, p. 143). This finding illustrates a serious gap that needs to be further investigated in terms of why there is no PMS in these municipalities and why the majority of employees do not have a performance agreement.

The mean score value is very low in the sub-dimension that all the employees in these municipalities have a performance agreement. On a scale from 1 to 5, the respondents were between 2,7097 and the standard deviation was 1,14920. This indicates that a high proportion of municipal employees ranged from 'agreed' to 'undecided' on statements relating to this sub-dimension. Much room for improvement is reflected for this sub-dimension, and certain employees at the present time may not have a performance agreement in the institution because of the dysfunctional PMS.

This finding on theme that the employees have a performance agreement in two selected municipalities. A male senior manager in the Wellness Centre Department in the BCMM had this to say:

I don't want to lie; the performance management system is not yet in place in this institution. The performance management system is only at the level of the heads of directorates and that is mainly due to the legislation that enforces the employer to do so. So mainly the people that are measured in terms of their performance are the HODs and the city manager, and they have performance agreements as a result of that.

The PMS is meant to counteract the legacy of poor-performing institutions, particularly to reduce the occurrence of poor service delivery that currently characterises South Africa's local government (Westhuizen, 2016). This finding suggests that South African municipalities need to innovate and learn how to continue with vital practices such as M&E in the new restrictive conditions (Fraser-Moleketi, 2020).

Table 1 shows a low percentage - 37% - of respondents who agreed that the performance agreement specifies the objectives and targets that should be met during COVID-19, against a close 35% who disagreed. Only 28% were undecided. This finding again indicates the negative impression of the absence of PMS in these municipalities. It also resonates with the study conducted by Naidoo (2012), who warned that the inadequate preparation of assessment plans and a lack of strong policy standards for M&E outcomes can yield poor results and recommendations. This may ultimately have an impact on decision-making processes based on facts. This finding proves that it is difficult to achieve municipal objectives and targets if PMS is not in existence, especially during the COVID-19 pandemic.

The mean score value is very low in the sub-dimension of the performance agreement specifying the objectives and targets that should be met during COVID-19 in these municipalities. The results showed that on a scale from 1 to 5, the respondents were between 2,9409 and the standard deviation was 1,17715. This indicates that a high proportion of municipal employees ranged from 'agreed' to 'undecided' on statements relating to this sub-dimension. This finding shows there is significant room for improvement and that these municipalities need to go back to basics when they are busy with planning. Again, this finding was confirmed by respondents who were interviewed qualitatively on the theme whether the performance agreement specifies the objectives and targets that should be met during COVID-19. A male senior manager in the Wellness Centre Department in the BCMM had this to say:

Well, that goes now to the absence of the employee performance management system, which [has] not been in place. However, having said that, as part of supervision and management, that is just an assumption because it's not something that [should] be done, but it would have been expected that the manager would be able to see if one is performing or not performing because the person is given a job description and duties to do.

This finding implies that the failures of the performance agreement to specify the objectives and targets that needed to be met during COVID-19 emanate from the absence of a PMS. These municipalities are inconsistent in terms of implementing a PMS system and are resistant to change. Consequently, an effective M&E tool would ensure the long-term sustainability of a government that is responsive to the needs and demands of communities (Mlambo & Masuku, 2020).

Table 1 shows the highest percentage of 43% of respondents who disagreed that time frames were set continuously to monitor their work during the COVID-19 crisis, against 38% who agreed and only 18% who were undecided. This finding resonates with the study by Mlambo and Masuku (2020) who concur that the COVID-19 pandemic has also unmasked poor leadership and governance, and the inability of the government to provide a high standard of performance. The mean score value in the sub-dimension that time frames were set continuously to monitor employees' work during the COVID-19 crisis in these municipalities revealed that on a scale from 1 to 5, the respondents were between 2,8333 and the standard deviation was 1,29830. This indicates that a high proportion of municipal employees disagreed with the statements relating to this sub-dimension. This finding leaves a significant gap for these municipalities. It is disconcerting and illustrates the serious gap that needs to be further investigated regarding why time frames are not set in these municipalities.

A female senior manager in the M&E Department at the BCMM confirmed this finding by saying:

In some cases, most of the KPIs of the performance information was compromised because there was no movement because people were on lockdown, it was really bad. In terms of time frames, in terms of content, in terms of the personnel available, it was bad. It's just the availability and adhering to time frames, timelines. Because remember, when they feed us with information, they get it from project managers. So basically the moment you don't get information, one could delay, it delays the whole report.

These findings confirm and emphasise the negative perceptions of employees regarding time frames being continuously set to monitor their work during the COVID-19 crisis. These municipalities need to consider alternate approaches to continue M&E operations during epidemics so that the ability of M&E to redirect government mitigation efforts is effective (Gaskin-Reyes, 2016).

Table 1 shows a very low percentage of respondents – 41% – who agreed that employees' work is evaluated whenever there is an early indication of poor performance. This was against 34% who disagreed and 26% who were undecided. Despite the higher percentage of respondents who agreed that their work is evaluated whenever there is an early indication of poor performance, the number of respondents who disagreed or were undecided is cause for concern. This finding is consistent with the study conducted by Matsiliza (2019) who concluded that South African municipalities rely on consultants and the Department of Cooperative Governance and Traditional Affairs to perform evaluations. As a result, Matsiliza (2019) further maintained that M&E employees and consultants have limited experience to support evaluations. The mean score value in the sub-dimension of employees' work being evaluated in these municipalities whenever there is an early indication of poor performance showed that on a scale from 1 to 5, the respondents were between 2,9570 and the standard deviation was 1,18014. This indicates that a high proportion of municipal employees ranged from 'agreed' to 'undecided' on statements relating to this sub-dimension. This finding is problematic as evaluations create intervention measures for municipalities to properly manage and to achieve service delivery mandates, which includes operations during COVID-19 (Ile et al., 2019).

The current study noted that there are some contradictions among senior managers on the theme regarding the work of municipal officials being evaluated when there is an early indication of underperformance. A male senior manager in the Disaster Management Department at the BCMM confirmed the following: 'Yes, the strategy is available, because due to quarterly reports, that's where they

see that there is a problem in certain areas because our reports highlight as such. What they do about these reports is a discussion for another day.' A dissenting argument from a participant from the M&E Department at the BCMM opined: 'If "I can be specific. [On] the issue of evaluation, as much as we talk about it, no manager really does reflect [it] in his/her department ... So there is a gap".

This study finding partly contradicts the study conducted by Ile et al. (2019) who affirmed that evaluations can be useful for illuminating performance problems, which then allow for the creation of genuine solutions. Indeed, as a rule, whenever there is a gap between expected results and realised results of a project, evaluation becomes mandatory (Chaplowe & Cousins, 2016).

Table 1 shows a high percentage of 42% of respondents who agreed that monitoring and evaluation are concepts for the top managers only, against 36% who disagreed and only 22% who were undecided. This finding is contrary to the study conducted by Mapitsa and Korth (2017) who argued that to properly comprehend the usefulness of the M&E system, the intent of such a system must be openly shared with all employees. This finding shows a significant need in terms of changing the perceptions of employees in these municipalities. The mean score value in the sub-dimension was 3.0484 and the standard deviation was 1,28729. This reveals that a high proportion of municipal employees ranged from 'agreed' to 'undecided' on statements relating to this sub-dimension. These averages reflect the current situation at the municipalities concerned – that there are differing views regarding monitoring and evaluation being concepts for the top managers only. Even though there is not much conclusive or empirical evidence on this sub-dimension, the study conducted by Kusek and Rist (2004) emphasised that managers should be the driving force behind the M&E system while at the same time ensuring the provision of material and human resources to make the system work.

The theme that emerged on monitoring and evaluation in these municipalities are still concepts for the top managers. A male senior manager from the Disaster Management Department at the NMBM gave this perspective: 'I would say there could be some improvement where people could be better informed as to M&E ... this should be rolled out and implemented on a more junior level as well, not just at senior management level'. This finding is consistent with earlier research by Kadel et al. (2020) who noted that M&E must also maintain a clear link with the decision-making system at various levels so as to delegate roles and jointly resolve challenges.

Table 1 shows a highly disproportionate 78% of respondents who agreed that M&E is very important. A total of 12% disagreed and 14% were undecided. The level of agreement was higher for this sub-dimension, indicating that more staff value M&E as being very important. Employees overwhelmingly had an understanding of what is M&E is. This sub-dimension indicates a high mean score value of 3,9839 which shows the respondents are in support of the statement that M&E is very important. The standard deviation was 1,18311. This result supports the findings of Sebake and Mukonza (2020) who consider M&E to be a tool to monitor projects and evaluate service delivery successes and failures in South African municipalities.

The identified theme on the importance of M&E in the two municipalities under investigation. Through interviews with senior managers from both municipalities, a unanimous acknowledgment of the crucial role of M&E emerged. These findings give insight on the pivotal nature of effective M&E practices, particularly in the midst of a crisis such as the COVID-19 pandemic. The recognition of M&E's importance extends beyond conventional viewpoints, providing a comprehensive understanding of the crisis. This holistic perspective encompasses various dimensions, including health-related aspects like disease management, the effectiveness of healthcare facilities, and the level of community engagement. In essence, the theme emphasizes the integral role that M&E plays in navigating and understanding the complexities of crises, contributing to a more informed and strategic response. In addition, Makwembere and Garidzirai (2021) contends that without regular M&E, which is the situation in most municipalities, irregularities are not detected and proper corrections cannot be timeously implemented. Monitoring and evaluation is only important if management decisions are backed up by accurate facts (Kadel et al., 2020).

6. Conclusion

The dimension of investigating the effectiveness of M&E strategies during COVID-19 was established to address the objective of the study. The findings show significant room for improvement in these two municipalities: NMBM and BCMM. The estimated number of confirmed COVID-19 mortalities stands at 2,354 in BCMM, while NMBM has reported an excess of 5,104 deaths (Bradshaw et al., 2021). The findings of the study show a low confidence of employees agreeing that COVID-19 was handled well in these municipalities. This implies that there were ineffective M&E strategies and an absence of organisational support to sustain best practice in these two organisations. Consequently, the absence of M&E strategies led to new corruption trends in South African municipalities during the COVID-19 pandemic.

Municipal employees are not convinced that they fully understand the M&E system. This suggests that they are not equipped to efficiently understand M&E systems. It also implies that there has been no proactive action taken in these municipalities in dealing with control weaknesses, even before the COVID-19 pandemic. In addition, employees are not convinced that communication about lockdown rules and regulations were clear in these municipalities. The research suggests that the communication channels were poor or inadequate, something that has encouraged employees to work in silos and which has had an adverse impact on the COVID-19 contingency plans.

The study findings substantiate the fact that not all municipal employees know that there are adequate policies to support the M&E system. It is noteworthy to conclude that the available policies were not aligned or reviewed to suit or respond to the current pandemic. Furthermore, municipal employees disagree that their work has been monitored during COVID-19. This demonstrates that performance management during the pandemic was absent and that there was no accountability and consequence management. It can be concluded that the line managers are in dereliction of their duties. Instead, they spent their time managing emotions rather than employees' performance, which paralysed service delivery. Similarly, employees were not sure that the PMS is used as an M&E tool. This finding suggests that M&E tools were not formally introduced in these municipalities, giving rise to a culture of poor performance, which continues to persist.

Employees of these two municipalities did agree that stakeholders' involvement was inadequate in the COVID-19 response plans. This finding implies that stakeholders are unwilling to engage in municipal business and have lost trust in these municipalities. This came about as a result of a lack of transparency, financial mismanagement and reports of corruption during COVID-19, which have been widely publicised in the media. Moreover, the study finds that employees disagree that all workers have performance agreements. The latter finding demonstrates that staff are not productive and there are no consequences for poor performance because they are not held accountable for their performance. Although the performance agreement does not necessarily lead to performance, but it helps to establish clear expectations, goals, and a framework for accountability within an organization.

There are differing views on the sub-dimension that a performance agreement specifies the objectives and targets that should be met during COVID-19. Surprisingly, there was appalling disagreement on time frames being continuously set to monitor work during the COVID-19 crisis. Again, municipal employees disagree that their work is evaluated whenever there is an early indication of poor performance. In these municipalities, M&E is perceived as being a concept for top managers only. Nevertheless, the majority of employees value the importance of M&E.

The bleak findings presented on this dimension clearly depict a linear relationship between municipalities' leadership failure to implement the PMS among its workforce and an exponential increase in COVID-19 cases and fatalities in the only two metros within the municipality. This highlights the critical importance of effective PMS in mitigating the impact of public health crises and the need for urgent interventions in municipal leadership practices to enhance pandemic response and outcomes. The empirical finding reiterates the fact that municipal employees are in the dark concerning the PMS. The major findings also show that municipal resources have been abused and that employees have unduly benefitted. The study indicates that municipal employees were still receiving car allowances even though these vehicles have not been driven for work purposes for almost a year. Staff with comorbidities have not been handled well. The findings also highlight that employees such as cleaners and security staff have been at risk of being infected by COVID-19 because they have not been given the necessary support in these municipalities.

7. Recommendations

Taking the findings of the present study into consideration, the study recommends that human resource managers design performance agreements that are aligned with the key indicators of the municipality

and which are relevant to disasters such as the COVID-19 pandemic. There was no suitable PMS during the disaster; accordingly, human resources led by a specialist should design a PMS that would be effective during disasters such as COVID-19. Employees in different employment categories, including managers, need to be trained on best practice of M&E tools in order to apply this during health-related disasters such as COVID-19. The municipalities need to develop strategies for communication change management. Each municipal department should devise a communication plan that is adjusted on a regular basis to suit employees' needs. The municipalities' policy and planning department should review all policies to respond appropriately to the pandemic and include all other forms of disasters such as floods, storm surges, wildfires and shack-fires, heatwaves, drought and water-shortages, waterborne diseases such as cholera and zoonotic diseases such as malaria etc. This unit should devise municipal by-laws to support the M&E system in order to curb the spread of the virus and other threats.

Municipal line managers should be trained on how to monitor employees - during global pandemics in particular – when people are working virtually. The Human Resource Department and line managers should revisit and revise performance agreements annually to be aligned with the current needs and demands as they relate to COVID-19. The department should also adopt digital tools in order to manage people working from home during this time of COVID-19. These municipalities' policies have not yet been updated to accommodate the pandemic. It is recommended that policies be updated and aligned with COVID-19 so as to ensure compliance with regulations and protocols. For example, because COVID-19 is unprecedented, the human resource development policy on training has been silent. On the subject of social distancing, training – either online or in person – would have an effect on groups as training is often attended by groups of individuals. Other policies may include the HIV/AIDS policy, while the employee wellness policy should also be updated. It is thus essential that municipal M&E and performance management be appropriately designed.

8. Limitations of the study

The limited focus of this study deals with two municipalities; their strategies therefore need to be considered in the broader approach adopted by the South African Government in fighting the COVID-19 pandemic. This article deals with the unfolding of the pandemic up until the end of June 2021. Anyone using this article as a case study needs to consider the study's findings in the context of the rapidly unfolding pandemic. When used in a learning context, relevant subsequent events in the COVID-19 pandemic need to be considered, as suggested in the paragraph that follows on recommendations for educational use. Future researchers should use different research methodologies to investigate the type of strategies applied in these municipalities during disasters. The study recommends that the current research be extended by future researchers and be undertaken in other municipalities around the country.

Disclosure statement

The authors declare that there are no potential conflicts of interest.

About the authors

Dr. Babalo Yekani, PHD in Public Management, Walter Sisulu University. Research interest: Monitoring and Evaluation.

Prof Bethuel Sibongiseni Ngcamu, PHD in Education (Higher Education), DTECH in Public Management, University of South Africa. Research interest: disaster risk management.

Dr Sareesha Pillay, PHD in Public Management, Nelson Mandela University. Research interest: service delivery.

ORCID

References

- African Development Bank. (2009). Building capacity for monitoring and evaluation in Africa: AFDB role and experience. Retrieved December 12, 2020, from http://www.nec2015.net/sites/default/files/AfDB%20Brief%20-%20Role%20 and%20Experience.pdf
- Awasthi, S. (2020). A robust monitoring & evaluation system can prove effective during pandemics. Retrieved July 24, 2020, from https://www.outlookindia.com/website/story/opinion-a-robust-monitoringevaluation-system-can-prove-effect ive-during-pandemics/35470
- Babbie, E., & Mouton, J. (2001). The practice of social research. Oxford University Press.
- Ball, I. (2020). Reflections on public financial management in the Covid-19 pandemic. Journal of Accounting & Organizational Change, 16(4), 1–18. https://doi.org/10.1108/JAOC-10-2020-0160
- Bandalos, D. L. (2018). Measurement theory and applications for the social sciences. Guilford Press.
- Bell, S., & Aggleton, P. (2016). Monitoring and evaluation in health and social development: Interpretive and ethnographic perspectives. Routledge.
- Boadu, E., & Ile, I. (2017). Rethinking participation in monitoring and evaluation: Beneficiaries' perspectives from Local Enterprises and Skills Development Programme (LESDEP) in Ghana. Loyola Journal of Social Sciences, 31(2), 209-223.
- Bradshaw, D., Laubscher, R., Dorrington, R., Groenewald, P., & Moultrie, T. (2021). Report on weekly deaths in South Africa 6 - 12 June 2021 (week 23). Burden of Disease Research Unit South African Medical Research Council 15 June 2021. UCT Centre for Actuarial Research. Retrieved February 28, 2024, from https://www.samrc.ac.za/sites/default/ files/bod/weeklyreports/weekly12June2021.pdf
- Chaplowe, S. G., & Cousins, J. B. (2016). Monitoring and evaluation training: A systematic approach. SAGE Publishing. Charmaz, K. (2009). Grounded theory: The SAGE encyclopedia of social science research methods. SAGE Publishing.
- Corburn, J., Vlahov, D., Mberu, B., Riley, L., Caiaffa, W. T., Rashid, S. F., Ko, A., Patel, S., Jukur, S., Martínez-Herrera, E., Jayasinghe, S., Agarwal, S., Nguendo-Yongsi, B., Weru, J., Ouma, S., Edmundo, K., Oni, T., & Ayad, H. (2020). Slum health: Arresting COVID-19 and improving well-being in urban informal settlements. Journal of Urban Health, 97(3), 348–357. https://doi.org/10.1007/s11524-020-00438-6
- Deji, F. (2019). What is a transformative design research framework? Retrieved June 25, 2020, from http://juliedebaveye. com/what-is-a-transformative-design-research-framework
- Den Hartog, D. N., Boselie, P., & Paauwe, J. (2004), Performance management: A model and research agenda, Applied Psychology, 53(4), 556-569. https://doi.org/10.1111/j.1464-0597.2004.00188.x
- Dlamini, P. C., & Migiro, S. (2016). Performance monitoring and evaluation systems in the South African local government. Journal of Public Administration, 51(3), 376-389.
- Draai, E., & Zazi, L. (2021). Performance management for responsive local governance: The case of Chris Hani District Municipality. Administratio Publica, 29(3), 89-107.
- Dube, K., Nhamo, G., & Chikodzi, D. (2021). COVID-19 pandemic and prospects for recovery of the global aviation industry. Journal of Air Transport Management, 92, 102022.
- Eresia-Eke, C. E., & Boadu, E. S. (2019). Monitoring and evaluation preparedness of public sector institutions in South Africa. Journal of Reviews on Global Economics, 8, 532-542. https://doi.org/10.6000/1929-7092.2019.08.46
- Evans, A. (2020). A pivotal year for monitoring and evaluation: Building and improving monitoring and evaluation capacity worldwide is key to achieving the Sustainable Development Goals within ten years. Retrieved October 12, 2020, from https://ieg.worldbankgroup.org/blog/pivotal-year-monitoring-and-evaluation
- Fouché, C. B., Strydom, H., & Roestenburg, W. J. H. (2022). Research at grass roots: For the social sciences and human services professions (5th ed.). Van Schaik Publishers.
- Fraser, D. I., & Morkel, C. (2020). State of monitoring and evaluation in Anglophone Africa: Centre for Learning on Evaluation and Results in Anglophone Africa's reflections. African Evaluation Journal, 8(1), a505. https://doi. org/10.4102/aej.v8i1.505
- Fraser-Moleketi, G. J. (2020). Public service and COVID-19: The future implications. Journal of Public Administration, *55*(3), 265–269.
- Gaskin-Reyes, C. (2016). Monitoring and evaluation. In International encyclopedia of geography: People, the earth, environment and technology (pp. 1-4). Wiley Online Library. https://doi.org/10.1002/9781118786352.wbieg0085.pub2
- Govender, I. G. (2013). Monitoring and evaluation systems for enhancing governance and government legitimacy in South Africa. Journal of Public Administration, 48(4), 811–823.
- Guest, G., Namey, E., & Chen, M. (2020). A simple method to assess and report thematic saturation in qualitative research. PLoS One, 15(5), e0232076. https://doi.org/10.1371/journal.pone.0232076
- Gxabuza, F. G., & Nzewi, O. (2021). Findings on the role of cooperatives in local economic development in Buffalo City Metropolitan Municipality: A participatory monitoring and evaluation approach. Africa's Public Service Delivery and Performance Review, 9(1), a387. https://doi.org/10.4102/apsdpr.v9i1.387
- Ile, I. U., Eresia-Eke, C. E., & Allen-Ile, C. (2019). Monitoring and evaluation of policies, programmes and projects. Van
- Kadel, I. M., Ahmad, F., & Basnet, N. (2020). Rethinking monitoring and evaluation practices: Lessons from the COVID-19 pandemic. ICIMOD. Retrieved January 02, 2021, from https://www.icimod.org/article/rethinking-monitoring-and-evaluation-pra ctices-lessons-from-the-covid-19-pandemic



Kanyamuna, V., Kotzé, D. A., & Phiri, M. (2020). Monitoring and evaluation systems: The missing strand in the African transformational development agenda. World Journal of Social Sciences and Humanities, 5(3), 160-175.

Kanyamuna, V., Munalula, M., & Mulele, C. S. (2019). Monitoring and evaluation legislation in Zambia – gap analysis. International Journal of Humanities, Art and Social Studies, 4(1), 15–25.

Kanyamuna, V., & Phiri, M. (2019). Who said monitoring and evaluation is not rooted in firm theoretical foundations? A review of relevant literature. International Journal of Humanities, Art and Social Studies, 1(4), 1-23.

Kariuki, P., & Reddy, P. (2017). Operationalising an effective monitoring and evaluation system for local government: Considerations for best practice. African Evaluation Journal, 5(2), 1-8. https://doi.org/10.4102/aej.v5i2.240

Kettner, P. M., Moroney, R. M., & Martin, L. L. (2015). Designing and managing programs: An effectiveness-based approach. SAGE Publications.

Khambule, I. (2020). The effects of COVID-19 on the South African informal economy: Limits and pitfalls of government's response. Loyola Journal of Social Sciences, 34(1), 95-109.

Kusek, J. Z., & Rist, R. C. (2004). Ten steps to a results-based monitoring and evaluation system: A handbook for development practitioners. World Bank Publications.

Levin, D. E. (2017). Introduction. The Journal of Law, Medicine & Ethics, 45(1_suppl), 8–10. https://doi.org/10.1177/1073110517704387 Mahlalela, P. T., Blamey, R. C., Hart, N. C. G., & Reason, C. J. C. (2020). Drought in the Eastern Cape region of South Africa and trends in rainfall characteristics. Climate Dynamics, 55(9-10), 2743-2759. https://doi.org/10.1007/s00382-020-05413-0

Makwembere, S., & Garidzirai, R. (2021), A case for ethical frameworks to enhance mobile government implementation at South African municipalities. The Journal for Transdisciplinary Research in Southern Africa, 17(1), a826. https:// doi.org/10.4102/td.v17i1.826

Malefane, S. (2009). Structuring South African municipalities for effective local economic development (LED) implementation. Journal of Public Administration, 44(1/1), 156-169.

Mantzaris, E., & Ngcamu, B. S. (2020b). Municipal corruption in the era of a COVID-19 pandemic: Four South African case studies. Journal of Public Administration, 55(3/1), 461-479.

Mapitsa, C. B., & Korth, M. T. (2017). Designing diagnostics in complexity: Measuring technical and contextual aspects in monitoring and evaluation systems. African Evaluation Journal, 5(1), a196. https://doi.org/10.4102/aej.v5i1.195

Matsiliza, N. S. (2019). Strategies to improve capacity for policy monitoring and evaluation in the public sector. Journal of Reviews on Global Economics, 8, 490-499. https://doi.org/10.6000/1929-7092.2019.08.42

Mello, D. M. (2018). Monitoring and evaluation: The missing link in South African municipalities. The Journal for Transdisciplinary Research in South Africa, 14(1), a409. https://doi.org/10.4102/td.v14i1.409

Mkama, W. (2017). Evolution of natural resource programs performance management in Tanzania: A review of monitoring, evaluation and learning (MEL) aspects and policy recommendations. Tanzania Natural Resource Forum. Retrieved February 21, 2021, from https://www.tnrf.org/files/pubs/project_performance_mgt_tanzania.pdf

Mlambo, V. H., & Masuku, M. M. (2020). Corruption and COVID-19: The final nail in the coffin for South Africa's dwindling public finances. Journal of Public Administration, 55(3/1), 449–587.

Mleke, M. N. (2020). A survey of monitoring and evaluation systems for government projects in Tanzania: A case of health projects. International Journal of Information Engineering and Electronic Business, 12(1), 8-18. https://doi. org/10.5815/ijieeb.2020.01.02

Moore, D. S., Notz, W. I., & Flinger, M. A. (2013). The basic practice of statistics (6th ed.). W.H. Freeman and Company. Motingoe, R. S., & Van der Waldt, G. (2013). A monitoring and evaluation system utilisation model for support to South African municipalities. Politiea: South African Journal for Political Science and Public Administration, 32(3), 4–29.

Mthethwa, R. M., & Jili, N. N. (2016). Challenges in implementing monitoring and evaluation: The case of the Mfolozi Municipality. African Journal of Public Affairs, 9(4), 102-113.

Munzhedzi, P. H., & Makwembere, S. (2019). Good governance as a solution to local economic development challenges in South African municipalities. Journal of Public Administration, 54(4/1), 659-676.

Naidoo, R. (2012). Subduing local voice – public participation and ward committees. In S. Booysen (Ed.), Local elections in South Africa: Parties, people, politics (pp. 279-293). SUN Media.

National Treasury. (2020). MFMA Circular No. 102. Retrieved June 12, 2020, from https://www.tralac.org/documents/ resources/Covid-19/countries/3762-mfma-circular-102-emergencyprocurement-in-response-to-national-stat e-of-disaster-south a frica-with-annexes-and-amendments-national-treasury-5 may-2020/file.html

Netshakhuma, N. S. (2020). Assessment of a South Africa national consultative workshop on the Protection of Personal Information Act (POPIA). Global Knowledge, Memory and Communication, 69(1/2), 58-74. https://doi. org/10.1108/GKMC-02-2019-0026

Nggobe, F., Fourie, D. J., & Thsiyoyo, M. (2021). Performance management and accountability practices in South African municipalities: Critical considerations. Administratio Publica, 29(3), 108-128.

Ngwakwe, C. C. (2020). Analysis of budgetary impact of COVID-19 in South Africa: Implications beyond COVID. Journal of Public Administration, 55(3/1), 450-460.

Organisation for Economic Co-operation and Development. (2020). The territorial impact of COVID-19: Managing the crisis across levels of government. Retrieved June 20, 2020, from https://www.oecd.org/coronavirus/policy-responses/ the-territorial-impact-of-covid-19-managing-thecrisis-across-levels-of-government-d3e314e1

Pietersen, J. M. (2020). Possible implications of COVID-19 for the conditions of service in the public sector. Journal of Public Administration, 55(3/1), 435-449.

Prodanova, J., & Kocarev, L. (2021). Is job performance conditioned by work-from-home demands and resources? *Technology in Society, 66,* 101672. https://doi.org/10.1016/j.techsoc.2021.101672

Public Service Commission (PSC). (2008). Basic concepts in monitoring and evaluation. PSC.

Quad, A. D. J. P. (2016). *Research tools: Interviews & questionnaires*. Research Methodology in Education. Retrieved October 20, 2021, from https://lled500.trubox.ca/2016/225

Quinn, G. P., & Keough, M. K. (2002). Experimental design and data analysis for biologists. Cambridge University Press. Republic of South Africa. (2002). Disaster Management Act, 2002 (Act No. 57 of 2002). Government Printers.

Republic of South Africa. (2013). The Protection of Personal Information Act, 2013 (Act No. 41 of 2013). Government Printers.

Republic of South Africa. (2020). *Treasury on coronavirus COVID-19 support for municipalities*. Retrieved June 20, 2020, from https://www.gov.za/speeches/treasury-coronavirus-covid-19-support-municipalities-12-may-2020-0000

Rosenberg, E., & Kotschy, K. (2020). Monitoring and evaluation in a changing world: A Southern African perspective on the skills needed for a new approach. *African Evaluation Journal*, 8(1), a472. https://doi.org/10.4102/aej.v8i1.472 Seasons, M. (2003). Monitoring and evaluation in municipal planning: Considering the realities. *Journal of the American Planning Association*, 69(4), 430–440. https://doi.org/10.1080/01944360308976329

Sebake, B. K., & Mukonza, R. M. (2020). Integrated development plan, monitoring and evaluation in the City of Tshwane: A confluence question for optimising service delivery. *Journal of Public Administration*, *55*(3), 342–352.

Sekaran, U. (2002). Research methods for business: A skill-building approach (4th ed.). John Wiley & Sons.

Sibanda, M. M., Zindi, B., & Maramura, T. C. (2020). Control and accountability in supply chain management: Evidence from a South African metropolitan municipality. *Cogent Business & Management*, 7(1), 1785105. https://doi.org/10. 1080/23311975.2020.1785105

South African Local Government Association (SALGA). (2020). Local government's response to Covid-19 Portfolio Committee: COGTA. Retrieved May 02, 2020, from https://www.salga.org.za/Municipalities%20MAO.html

Taber, K. S. (2018). The use of Cronbach's alpha when developing and reporting research instruments in science education. *Research in Science Education*, 48(6), 1273–1296. https://doi.org/10.1007/s11165-016-9602-2

Westhuizen, E. J. (2016). Human resource management in government. A South African perspective on theories, politics and processes. Juta & Co.

World Health Organization. (2020). Considerations for mass gatherings in the context of COVID-19: Annex: Considerations in adjusting public health and social measures in the context of COVID-19. Retrieved December 20, 2020, from https://www.who.int/publications/i/item/considerations-for-mass-gatherings-in-the-context-ofcovid-19-annex-considerations-in-adjusting-public-health-and-social-measures-in-the-contextof-covid-19

Xue, Y., Turner, J. R., Lecoeuvre, L., & Anbari, F. (2013). Using results-based monitoring and evaluation to deliver results on key infrastructure projects in China. *Global Business Perspectives*, 1(2), 85–105. https://doi.org/10.1007/s40196-012-0002-3